Author’s response to reviews

Title: Development of anti-glomerular basement membrane glomerulonephritis during the course of IgA nephropathy: a case report

Authors:

Tadasu Kojima (kojima802@gmail.com)
Go Hirose (joltexpert@yahoo.co.jp)
Shuhei Komatsu (moyashi1986@gmail.com)
Taito Oshima (fantasia.taito@gmail.com)
Kentaro Sugisaki (kensugisaki2000@yahoo.co.jp)
Tomohiro Tomiyasu (tomitomo1101@yahoo.co.jp)
Noriko Yoshikawa (lento0320@yahoo.co.jp)
Muneharu Yamada (myamada@tokyo-med.ac.jp)
Takashi Oda (takashio@tokyo-med.ac.jp)

Version: 3 Date: 21 Nov 2018

Author’s response to reviews:

November 21, 2018

Professor Jing Zhang

Editor

BMC Nephrology

Re: Manuscript BNEP-D-17-00582 R2

Dear Professor Zhang,
Thank you very much for your letter dated November 15, 2018, regarding our manuscript entitled “Development of anti-glomerular basement membrane glomerulonephritis during the course of IgA nephropathy: a case report”.

We have revised the manuscript in accordance with your directions, and are submitting a revised version of the manuscript. This time we upload our manuscript as a final, clean version following your direction. We have also attached a list of itemized responses to your directions.

We hope that we have responded adequately to your directions and that you will find our revised manuscript acceptable for publication in BMC Nephrology. If you have any questions, please contact me at the E-mail address shown below. Thank you very much for your kind consideration and we look forward to hearing from you.

Sincerely,

Takashi Oda, M.D., Ph.D.

Director, Kidney Disease Center
Professor, Department of Nephrology and Blood Purification
Tokyo Medical University Hachioji Medical Center
1163 Tatemachi, Hachioji-shi, Tokyo 193-0998, Japan
Tel.: +81-42-665-5611
Fax: +81-42-665-1796
E-mail: takashio@tokyo-med.ac.jp

Responses to the Editor’s comments
1. **Overlap:** Please re-phrase these sections to minimize overlap.

Response:

Thank you for the valuable advice. Actually, however, the report that you indicated as the textual overlap was the abstract of our case presentation at the 18th International Vasculitis & ANCA Workshop in 2017. Therefore, we clearly mentioned this fact in the Acknowledgements section. (page 11, lines 6-8).

2. **Declaration:** Please include the following “Ethical approval and consent to participate: not applicable”

Response:

In accordance with your suggestion, we included the declaration.

3. **Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.**

Response:

We uploaded clean version of our manuscript and figures/tables as indicated.